ACE Weekly Report:  05-03-00

All ACE spacecraft subsystems are performing nominally. Attitude maneuvers were successfully executed on 04-30-00 and 05-03-00.  The next attitude maneuver is scheduled for 05-09-00.

On 04-27-00 AOS was delayed ~11 minutes due to problems with Station 66’s TGC (DR# L04235). Telemetry received in MOC after a second TGC was configured and placed online.  All data successfully captured during the support.

On 04-28-00 ACE did not have a DSN support. D24 was unable to support ACE due to a power problem at SPC-10 (DR# G05936).  There were no other stations available.  Two additional hours were appended to the next day’s support (day 120) to allow for all SSR data to be captured  Both NOAA and USAF were advised of the changes to support times.  All SSR data was successfully recovered during the next support on day 120.  An attitude maneuver had been originally scheduled for the Day 119 support.  In order for the HGA to not exceed its constraint, the maneuver would need to be performed prior to Monday’s support.  FDF agreed to perform the maneuver on Sunday, day 121.

On 05-02-00 the clock calibration file became corrupted.  As a result the spacecraft time was being incorrectly converted.  The backup calibration file was copied, TPOCC recycled and the time was now valid.  The support proceeded without any further problems (AR# G00-0016).  On 05-03-00 the clock calibration file was again corrupted.  After consulting with the developer it was decided to issue the TPOCC setepoch directive.  The directive provides the clock calibration software with the distance to the ACE spacecraft.  The software uses the range to compensate for transmission delays between the spacecraft ticks and the Ground Receipt Time (GRT).  Once the directive was issued, the spacecraft time was valid and correct (AR# G00-0017).  The support proceeded nominally.  The cause of the corruption is unknown;  the developer is currently investigating.

The SWICS Instrument team is adjusting the voltage of their Post-Accelerator Power Supply (PAPS).  It will take several sessions to complete the process (OCR SWX-0020).  Session 1 was performed on 04-27-00, Session 2 on 05-01-00 and Session 3 was performed on 05-03-00.

ULEIS Instrument Status:

No commanding to the ULEIS instrument has been performed this week.  The ULEIS instrument team is continuing their investigation of the anomaly.  No new information is available at this time.
